Author's response to reviews

Title: Smoking, alcohol consumption, physical activity, and family history and the risks of acute myocardial infarction and unstable angina pectoris: a prospective cohort study

Authors:

Audrey HH Merry (Audrey.Merry@epid.unimaas.nl)
Jolanda MA Boer (Jolanda.Boer@rivm.nl)
Leo J Schouten (LJ.Schouten@epid.unimaas.nl)
Edith JM Feskens (Edith.Feskens@wur.nl)
WM Monique Verschuren (Monique.Verschuren@rivm.nl)
Anton PM Gorgels (t.gorgels@mumc.nl)
Piet A van den Brandt (PA.vandenBrandt@epid.unimaas.nl)

Version: 7 Date: 29 November 2010

Author's response to reviews:

Dear editor,

Some small changes are made in the final version of the manuscript.